Author's response to reviews

Title: Brain imaging in Kufs disease type B. Case report

Authors:

Roberto Di Fabio (rodifa@gmail.com)
Claudio Colonnese (claudio.colonnese@uniroma1.it)
Filippo M Santorelli (filippo3364@gmail.com)
Liliana Pestillo (pestilloliliana@virgilio.it)
Francesco Pierelli (francesco.pierelli@uniroma1.it)

Version: 2
Date: 6 May 2015

Author's response to reviews:

To Prof Timothy Shipley 6.5.2015
Editor-in-Chief of BMC Neurology

Dear Editor,

Please find enclosed a manuscript entitled “Brain imaging in Kufs disease type B. Case report” written by Drs Di Fabio, Colonnese, Santorelli, Pestillo, Pierelli, that we would like to submit for publication in the BMC Neurology as case report.

The authors declare that:
- The content of this manuscript have not been copyrighted or published previously and is not now under consideration for publication elsewhere.
- The corresponding author takes full responsibility for the data, the analyses and interpretation, and the conduct of the research; has full access to all of the data, and has the right to publish any and all data separate and apart from any sponsor.
- The corresponding author has received consent forms from any participant in the study, has them on file in case they are requested by the editor and has received an authorization-for-disclosure form for the imagines of the participants.
- All authors of this paper have no competing interests and approved the final version of the manuscript.
- Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal.

Thank you for your consideration of this manuscript. We look forward to hearing from you.

Yours sincerely,